Evaluating the applicability of a screen diffraction approximation to local volcano infrasound

Sean P. Maher^{* α}, Robin S. Matoza^{α}, Catherine de Groot-Hedlin^{β}, Keehoon Kim^{γ}, Kent L. Gee^{δ}

^αDepartment of Earth Science and Earth Research Institute, University of California, Santa Barbara, CA, 93106, USA.

 eta Scripps Institution of Oceanography, University of California, San Diego, La Jolla CA, 92093, USA.

 $^{\gamma}$ Geophysical Monitoring Program, Lawrence Livermore National Laboratory, Livermore, California, USA.

 $^{\delta}$ Department of Physics and Astronomy, Brigham Young University, Provo, UT 84602, USA.

Abstract

Atmospheric acoustic waves from volcanoes at infrasonic frequencies ($\sim 0.01-20$ Hz) can be used to estimate important source parameters for hazard modeling, but signals are often distorted by wavefield interactions with topography, even at local recording distances (<15 km). We present new developments toward a simple empirical approach to estimate attenuation by topographic diffraction at reduced computational cost. We investigate the applicability of a thin screen diffraction relationship developed by Maekawa [1968, doi: https://doi.org/10.1016/0003-682X(68)90020-0]. We use a 2D axisymmetric finite-difference method to show that this relationship accurately predicts power losses for infrasound diffraction over an idealized kilometer-scale screen; thus providing a validation for the scaling to infrasonic wavelengths. However, the Maekawa relationship overestimates attenuation for realistic volcano topography (using Sakurajima Volcano as an example). The attenuating effect of diffraction may be counteracted by constructive interference of multiple reflections along concave volcano slopes. We conclude that the Maekawa relationship is insufficient as formulated for volcano infrasound, and suggest future modifications that may improve the prediction capability.

> Keywords: Volcano infrasound; Acoustic diffraction; Volcano topography; Attenuation; Acoustic focusing;

1 INTRODUCTION

Erupting volcanoes produce atmospheric acoustic waves dominant in the infrasonic frequency range $(\sim 0.01-20 \text{ Hz})$ that can be used to detect, locate, and characterize ongoing eruptions, and to estimate important eruption source parameters [e.g. De Angelis et al. 2019; Fee and Matoza 2013; Johnson and Ripepe 2011; Matoza et al. 2019]. However, outdoor propagation effects complicate direct interpretation of source processes. For example, recordings of volcano infrasound have been used to estimate volume flux [e.g. Iezzi et al. 2019a; Johnson and Miller 2014; Kim et al. 2015], vent radius [Muramatsu et al. 2018], and crater geometry [Johnson et al. 2018a; Johnson et al. 2018b], but wavefield interactions with topography lead to variable source estimates from signals recorded at different azimuths [e.g. Iezzi et al. 2019a; Kim and Lees 2014; Lacanna and Ripepe 2013; Maher et al. 2020; McKee et al. 2014].

In particular, diffraction over the crater rim and other topographic barriers [e.g. Kim and Lees 2011; Lacanna et al. 2014] distorts waveform character and reduces acoustic power compared to predictions based on spherical spreading over flat ground. The effect of diffraction over topography on acoustic power can be accounted for in full wavefield numerical simulations [Fee et al. 2017b; Iezzi et al. 2019b; Kim et al. 2015; Lacanna and Ripepe 2013; Matoza et al. 2009], but a rapid analytical approach would be useful in some scenarios. For example, rapid correction of observed signals is desirable when time constraints prohibit the use of numerical modeling, such as when planning station locations for the deployment of a temporary infrasound network during a rapid eruption response. It could also be advantageous when infrasound is generated by valley-confined surficial mass movements [e.g. Allstadt et al. 2018] or eruptions from a crater with rapidly evolving morphology [e.g. Ortiz et al. 2018].

The simplest expression for acoustic power loss to diffraction assumes a single point of diffraction over a thin screen, as proposed by Maekawa [1968]. This relationship was empirically derived from experiments in which attenuation was estimated as the difference in sound pressure levels behind a rigid screen and the predicted values for free field propagation. Later studies incorporated the thin screen approximation into expressions for thick barriers [Fujiwara et al. 1977] and wedges [Maekawa and Osaki 1985]. These and various other numerical approaches [e.g. Hadden and Pierce 1981; Pierce 1974] deal exclusively with audible-

^{*}Corresponding author: smaher@ucsb.edu

frequencies (20 Hz–20 kHz) and require complicated calculations and knowledge of the barrier geometry relative to the source and receiver.

The thin screen approximation has been discussed in the context of volcano infrasound by Lacanna and Ripepe [2013] and Ishii et al. [2020], but the accuracy of predictions made according to the Maekawa [1968] relationship have not been evaluated. Lacanna and Ripepe [2013] used a 2D finite-difference simulation to show that synthetic amplitudes recorded on a topographically-obstructed side of the crater rim of Stromboli Volcano are attenuated by 8 dB compared to those recorded at the same distance on a less topographically-obstructed side of the crater. Ishii et al. [2020] computed theoretical pressure fields in the vicinity of thin screen approximations to topography at Sakurajima Volcano, where the 2D topography profiles are rotated to match the elevations of source and receiver, then replaced by flat ground with a thin screen at the height and distance of the tallest obstruction. Ishii et al. [2020] calculate idealized pressure fields as a sum of two contour integrals representing geometrical and diffraction aspects for monopole radiation in cylindrical polar coordinates [Li and Wong 2005; Macdonald 1915].

The results of Ishii et al. [2020] suggest that accounting for the attenuation by the thin screen provided a better prediction of the relative amplitudes at several stations than a 1/r spreading correction alone. However, Ishii et al. [2020] did not compare their attenuation results to the empirical relationship proposed by Maekawa [1968], which could provide simpler and more rapid estimates than their analytical approach. Furthermore, the results of Ishii et al. [2020] neglect full-wavefield effects associated with realistic topography (e.g. focusing of reflections down concave slopes) and may therefore overestimate the true losses.

We seek to directly evaluate the applicability of the empirical relationship for thin screen diffraction [Maekawa 1968] to infrasonic wavelengths in general and to volcano acoustics in particular. We use numerical modeling to generate synthetic waveforms for full wavefield propagation over flat ground and multiple barrier types, including Sakurajima topography, to obtain attenuation estimates directly comparable to the Maekawa [1968] experiments. Our approach is conceptually similar to that of Le Pichon et al. [2012], who used large numerical simulations to derive a semiempirical predictive relationship for infrasound attenuation due to the influence of stratospheric winds on the acoustic wavefield during propagation over a global scale. Analogously, we use numerical simulations to provide progress towards a semi-empirical relationship for infrasound attenuation due to the effect of topographic diffraction on wavefield propagation at a local scale. Our study differs from Le Pichon et al. [2012] in that we use synthetic results to investigate the applicability of an existing empirical relationship rather than

PX

to propose a new empirical relationship. These results represent a first step towards the development of an accurate predictive relationship and directions for future research are discussed in Section 9.4.

This manuscript is organized as follows. Section 2 describes the thin screen diffraction approximation that is investigated here in the context of volcano infrasound. Section 3 describes Sakurajima Volcano and the observational dataset that is used in this study. Section 4 describes our primary methodology for numerically simulating infrasound propagation. Section 5 presents the main results of our simulations. Section 6 compares the results of our simulations to those from a different methodology that has been previously used with Sakurajima [Kim and Lees 2014]. Section 7 examines attenuation by diffraction in the frequency domain. Section 8 compares our synthetic results and thin screen predictions to observational data from Sakurajima. Section 9 discusses the implications of the study, limitations of our methodology, and directions for future work. Section 10 concludes the manuscript with the main findings and interpretations of the study.

2 Thin screen diffraction approximation

To first order, a barrier to acoustic propagation can be approximated by a screen with height and length much greater than its thickness. The acoustic shielding effect of a thin screen was quantified in a seminal study by Maekawa [1968], who investigated the efficiency of a thin screen to reduce sound pressure for general noise reduction applications (e.g. road noise). Maekawa [1968] recorded sound pressure behind a 2 m tall screen during a controlled indoor experiment and used the results to calculate the sound reduction compared to free field propagation. The shielding effect of the screen causes acoustic power losses during diffraction which can be related to the Fresnel number N [Pierce 1981]. The Fresnel number is the extra diffracted path length normalized by half a wavelength:

$$N = \frac{(R_{\rm t} - R_{\rm d})}{\lambda/2},\tag{1}$$

where R_t is the length of the diffracted path, R_d is the length of the direct path to a receiver (line-of-sight slant distance) and λ is wavelength. The orange and green lines over the volcano topography profiles in Figure 1 show example line-of-sight and diffracted paths, respectively, where the diffracted path is computed as the shortest possible source-receiver path over obstructions.

Maekawa [1968] presents an empirical chart of insertion loss as a function of Fresnel number which has become a standard reference for first-order noise reduction estimates in the field of audible acoustics [e.g. Ekici and Bougdah 2003; Hohenwarter 1990; Plotkin et al. 2009; Vu 2007]. Lacanna and Ripepe [2013] present a



Figure 1: Map of Sakurajima Volcano in southern Japan with 1 m resolution around the crater and 10 m resolution around the perimeter (interpolated to 1 m resolution). Contours are marked at 200 meters and the 1000 meter contour is marked in bold. The active vent is in Showa crater (upward triangle). Downward triangles represent infrasound sensors deployed for eight days in 2013 [Fee et al. 2014]. Profiles at right show topography between the vent and each sensor. Dash-dotted and solid lines show the direct and shortest diffracted raypaths, respectively, with lengths corresponding to R_d and R_t in Equation 1, respectively.

logarithmic fit to this chart which was originally introduced by Tatge [1973] and is commonly used in engineering applications:

$$IL_{pT} = 10\log(3 + 20N),$$
 (2)

where IL_{pT} is the "predicted" insertion loss, in decibels, with subscript "T" indicating the study [i.e. Tatge 1973]. This expression carries the advantage of fitting the entire range of Fresnel numbers considered in the chart; however, the accuracy of the fit is reduced at N < 1, because the abscissa is adjusted to make the fit a straight line in the range -0.3 < N < 1 [Maekawa 1968]. Note that negative Fresnel numbers indicate regions with direct line of sight to the source, but close enough to the shadow zone to feature influence of diffraction effects.

Since the source–receiver geometries at Sakurajima mainly feature N < 1, greater accuracy in this range is desirable. We therefore follow the approach of Ya-mamoto and Takagi [1992], who provide independent fits in the ranges -0.3 < N < 1 and $N \ge 1$, with a maximum deviation of 0.5 dB in the range -0.3 < N < 1:

$$IL_{\rm pY} = \begin{cases} 5+8|N|^{0.438} & \text{for } -0.3 < N < 1\\ 10\log_{10}(N) + 13 & \text{for } N \ge 1. \end{cases}$$
(3)

 $\langle\!|$ Presses universitaires de Strasbourg

It is important to note that while the Maekawa [1968] chart is commonly used as a first-order noise reduction estimate regardless of source-receiver geometry [e.g. Ekici and Bougdah 2003; Ettouney and Fricke 1973; Plotkin et al. 2009; Scholes et al. 1971; Vu 2007; Young et al. 2015], Maekawa [1968] states that it is intended for cases where the source and receivers are in the air. This condition allows direct comparison between diffracted measurements and free-field estimates since the influence of ground reflection can be avoided. However, in volcano infrasound studies, the sources and receivers are commonly on the ground. A chart relating N to attenuation for the case when sources and receivers are on the ground is provided by Fehr [1951] and can be fit by the equation:

$$IL_{\rm pF} = 10 + 10\log_{10}(N). \tag{4}$$

We find that Equation 4 generally predicts the attenuation for our synthetic volcano infrasound results better than Equations 2 and 3; however, the original paper by Fehr [1951] unfortunately does not provide explanation or justification for the relation. The chart is nonetheless reproduced in Figure 3 of Maekawa [1968] and was commonly cited before Maekawa's empirical relation became popular [e.g. Ettouney and Fricke 1973; Jonasson 1972; Purcell 1957; Scholes et al. 1971]. The predicted insertion loss given by Equations 2, 3, and 4 can be tested against experimental results described by modeled insertion loss (IL_m) [Lacanna and Ripepe 2013]:

$$IL_{\rm m} = -20\log\left(\frac{p_{\rm s}}{p_{\rm r}}\right),\tag{5}$$

where p_s is peak pressure behind an obstacle and p_r is peak pressure at the same distance without the obstacle. Here, p_s is peak pressure of the first wave cycle of a synthetic signal recorded behind a rigid obstacle using a finite-difference time-domain (FDTD) method for linear infrasound propagation [Section 4; de Groot-Hedlin 2017]. The synthetic peak pressure $p_{\rm r}$ is recorded at the same receiver location but with a flat lower boundary. We mainly use peak waveform pressures instead of Fourier amplitudes at a specific frequency because signal frequency content may vary inconsistently with different source processes such that a single frequency component may not be dominant in every event. However, in Section 8 we compare Fourier amplitudes at 0.5 Hz between observations, predictions, and synthetics to allow direct comparison to results from Ishii et al. [2020].

We hereafter refer to Equations 2, 3, and 4 as "predicted" insertion loss and to Equation 5 as "modeled" insertion loss in order to express the difference between predictions and our results from numerical modeling. We hypothesize that modeled insertion loss will be well approximated by predicted insertion loss if the attenuating effect of the barrier in question (e.g. wedge, Sakurajima topography) is similar to that of a thin screen.

3 SAKURAJIMA VOLCANO

Sakurajima is a highly active andesite–dacite stratovolcano in southern Kyushu, Japan, near the southern rim of the Aira Caldera (Figure 1). The 20 km³ edifice consists of lava flows and pumice fall deposits from at least 12 Plinian eruptions [Aramaki 1984]. From 1955 to present the eruptive activity has been characterized by ash-rich Vulcanian explosions [Ishihara 1985]. During the periods 1955–2006 and 2017–present, eruptions occurred from the summit vent of Minamidake, whereas from 2006 to 2017, eruptions occurred from a vent in Showa Crater on the southeast flank of Minamidake [Iguchi et al. 2013].

Due to the volcano's frequent activity (~1000 events per year since 2009) and proximity to the city of Kagoshima, Sakurajima is one of the world's bestinstrumented volcanoes [e.g. Iguchi et al. 2013] and a common target for infrasound studies. Acoustic waves generated by the explosive Vulcanian eruptions feature large amplitudes [449 Pa at 2.3 km; Fee et al. 2014; Matoza et al. 2014] and are sometimes visually observed as shock-like changes in luminance [Ishihara 1985; Yokoo and Ishihara 2007]. These studies and others [e.g.

arsigma Presses universitaires de Strasbourg

Aizawa et al. 2016; Cimarelli et al. 2016; Garcés et al. 1999; Iguchi et al. 2013; Kawakatsu et al. 1992; Maryanto et al. 2008; Miwa and Toramaru 2013; Morrissey et al. 2008; Smith et al. 2018; Tameguri et al. 2002; Uhira and Takeo 1994; Yokoo et al. 2009] illustrate the value of Sakurajima as an archetypal research target.

4 **FINITE-DIFFERENCE MODELING**

Our primary methodology involves numerical simulations of linear infrasound propagation with the FDTD code developed by de Groot-Hedlin [2017]. The model features a 2D cylindrical coordinate system in a sourcereceiver plane that is axisymmetric about the left boundary. This quasi-2D geometry models amplitude decreases with range identically to spherical spreading from a point source in 3D Cartesian space (1/r decay)without topography). In a 2D Cartesian model, the amplitude decrease with range would instead reflect a line source $(1/\sqrt{r} \text{ decay})$. References in this manuscript to "2D" modeling refer to the axisymmetric method of de Groot-Hedlin [2017] rather than a 2D Cartesian model. The simulations are advanced in time with finite differences over two time-steps using the modified midpoint method of Press et al. [1996], yielding second order accuracy in time of the solution. The program uses a staggered grid in which pressure is updated in cell centers while particle velocities are updated at cell boundaries. This staggered grid approach provides second order accuracy in the spatial derivates of the finite differences [Ostashev et al. 2005]. The lower boundary features rigid stair-step topography while the top and right boundaries feature absorbing perfectly matched layers [Berenger 1994].

The main results described in Section 5 are obtained from simulations performed in a 12×12 km model space with a constant sound speed of $340\,\mathrm{m\,s^{-1}}$ and an ambient density of 1.229 kgm⁻³. We require a minimum of 20 grid cells per wavelength at 5 Hz, yielding grid spacings of 3.4 m and time steps of 2.4 ms. The accuracy of synthetic frequency components is generally considered acceptable for discretization up to 10 grid nodes per wavelength [Taflove and Hagness 2005]. In these simulations the discretization criterion of 10 nodes per wavelength is met up to a frequency of 10 Hz, so synthetic results can be considered accurate up to 10 Hz. However, we chose to only interpret frequencies up to 5 Hz to maintain confidence in the numerical accuracy of the results. Note that in Section 6 we present simulations with a maximum source frequency of 1 Hz and a constant sound speed of $349 \,\mathrm{m\,s^{-1}}$ to facilitate direct comparison to 3D modeling by Kim and Lees [2014].

We run simulations with a variety of lower boundary conditions including flat ground, an idealized thin screen 800 m in height, (Figure 2, third row), a 1 km wide by 800 m tall wedge (Figure 2, fourth row), and 1 km by 800 m square barrier (Figure 2, fifth row). The height and distance of the thin screen is designed to scale the Maekawa [1968] experiment to infrasonic wavelengths and to create overlap in Fresnel numbers with the Sakurajima topography profiles. The geometries of the wedge and wide barrier are designed to create overlap in Fresnel numbers, facilitating direct comparison of the resulting attenuation to that of the screen and Sakurajima. We calculate Fresnel numbers for the synthetic stations assuming a wavelength of 340 m, corresponding to ~1 Hz peak frequency in the synthetic signals, and the homogeneous sound speed of 340 ms^{-1} .

For volcano topography we use Sakurajima Volcano as a case study because it is a common target for volcano infrasound research (see Section 3) and features rugged topography that complicates the acoustic wavefield [Johnson and Miller 2014; Kim and Lees 2014; Lacanna et al. 2014; McKee et al. 2014]. We use 2D topography profiles for each of five infrasound sensors deployed at Sakurajima in July 2013 [Fee et al. 2014] (Figure 1). We place the acoustic source inside Showa crater at the left boundary and propagate the wavefield over topographic profiles corresponding to the azimuths to the sensors in the 2013 deployment. For the lower boundary we use topography profiles from a 10m resolution digital elevation model (DEM) covering a broad area over Kagoshima Bay and a smaller 1-m resolution DEM covering the summit region and Showa crater. We fuse the two DEMs together after interpolating the 10-m resolution DEM to 1-m resolution over a rectangular mesh using a bivariate spline approximation. While our approach may be generalized to any topographic profile, the results presented here follow from previous work on the 2013 dataset [Fee et al. 2014; Johnson and Miller 2014; McKee et al. 2014; Yokoo et al. 2014] and may inform future studies that use the same data.

Synthetic peak pressures are recorded at 100 m spacing behind each obstruction (Figure 2) and compared to those recorded over flat ground to estimate IL_m (Equation 5). We use peak pressure to calculate IL_m such that the direct wavefront is measured rather than laterarriving reflections. Our approach allows us to thoroughly compare modeled insertion loss to predicted insertion loss by providing synthetic pressures at low computational cost for numerous source-receiver geometries representing a range of Fresnel numbers.

5 Results

The primary results of our study are the comparisons between predicted insertion loss from Maekawa's empirical relationship (Equation 3) and modeled insertion loss (Equation 5) based on the peak pressures of synthetics from 2D axisymmetric FDTD simulations. Fig-



ure 2 shows wavefield snapshots for five simulations. The compressional source is centered at an altitude of 660 m, the height of Showa crater. In the flat ground case, most acoustic energy is concentrated at the wavefront and the singular ground reflection behind it. In the Sakurajima KUR case, multiple reflections down the concave slope (see also Figure 1) lead to wavefield complexity and focusing, as also noted previously by Lacanna et al. [2014]. In the cases of the thin screen, wedge, and square barrier, diffraction reduces amplitudes behind the obstruction compared to the portions of the wavefield propagating higher than the obstruction. Reflections from the mirrored obstruction across the axisymmetric left boundary appear in the last one or two snapshots, but these phases do not influence the peak pressure insertion loss results because only the first arriving wave cycles are used in the analysis.

Figure 3A shows *IL*_m results all synthetic receivers over each lower boundary type. For 0.1 < N < 1 the thin screen results (red circles) show good agreement (<1 dB difference) with Equation 3 (black line). For the square barrier (cyan squares) and wedge (yellow triangles), the attenuation is underpredicted and overpredicted, respectively, by Maekawa's empirical relationship by a few dB, as expected from previous studies [e.g. Fujiwara et al. 1977; Maekawa and Osaki 1985; Pierce 1974]. Attenuation at synthetic Sakurajima receivers (diamonds in Figure 3A) is significantly overpredicted by Equation 3 and even negative at KUR at low Fresnel numbers (N < 0.2), indicating gains in power relative to the flat ground case, rather than losses, due to topographic focusing effects. The volcano results are better predicted by Equation 4 (grey line), which is formulated for sources and receivers on the ground, but which only covers N > 0.1.

As a guide, Figure 3B shows how the Fresnel number of each receiver varies with range (distance from the left boundary). Fresnel numbers generally decrease with distance as the difference between R_t and R_d decreases; however, at close ranges over the Sakurajima profiles, Fresnel numbers can increase with range as the propagation path becomes more obstructed by the crater rim.

6 Comparison of 2D cylindrical to 3D Cartesian modeling

To corroborate the results of our 2D numerical modeling in an axisymmetric geometry, we run additional simulations for direct comparison to previous 3D Cartesian finite-difference modeling with Sakurajima topography [Kim and Lees 2014]. As in the simulations presented by Kim and Lees [2014], we implement a homogeneous atmosphere with $349 \,\mathrm{m\,s^{-1}}$ sound speed, a maximum source frequency of 1 Hz with a discretization of 30 nodes per wavelength. Synthetics are recorded at the five stations of the July 2013 de-



Figure 2: Wavefield snapshots for infrasound propagation simulations over five lower boundary types. Each row represents a time sequence from left to right for one lower boundary. From top to bottom these lower boundaries are flat ground, Sakurajima topography (KUR profile), thin screen, wedge, and wide barrier, respectively. Downward triangles represent synthetic receivers. The full model space dimensions are 12×12 km and simulation run times are 34 seconds.

ployment with topography profiles as shown in Figure 1. We additionally run new 2D cylindrical and 3D Cartesian simulations that approximate flat-ground propagation for the same source-receiver distances so that attenuation by topography may be estimated using Equation 5. In the 2D cylindrical case, the flat ground simulation features a flat surface at sea level, with the source on the ground at the left boundary. Receivers are placed on the ground at distances corresponding to the slant distances (R_d) of the Sakurajima stations, measured from the edge of the source radius. We calculate Fresnel numbers for the synthetic stations assuming a wavelength of 873 m, corresponding to \sim 0.4 Hz peak frequency in the synthetic signals, and the sound speed 349 m s^{-1} . We approximate the source function of this simulation by placing a receiver 10 m outside the source radius. The source radius is 1386 m and the model dimensions are 12×12 km.

Both methods use a staggered grid scheme [Ostashev et al. 2005] such that synthetic pressures are recorded at grid cell centers. This means that receivers cannot be placed directly on the ground and are instead located 5 meters above the surface in these simulations. When the height of a receiver is large compared to its range and the acoustic wavelength this may introduce undesired interference between the direct and groundreflected waves which would be in-phase on the surface. This effect is largely insignificant for the wavelengths considered here and at receivers more than a few hundred meters from the source. However, in the 3D Cartesian case, the source function receiver is placed 10 meters from the point source and thus is subject to phase-shifting in the ground-reflected wave. For this reason we approximate flat ground propagation in the 3D Cartesian case by recording pressure in the free field and multiplying the amplitudes by two. This approach accurately models the constructive interference of the in-phase direct and ground-reflected waves in the flat ground case by accounting for the image sources inherent to flat ground propagation [Kim et al. 2012; Morse and Ingard 1986]. For the 2D cylindrical case, the source function receiver is placed at a significant



Figure 3: [A] Attenuation as a function of Fresnel number for synthetic peak pressures from 2D cylindrical simulations with multiple lower boundary types (screen, wedge, square, and five azimuths at Sakurajima Volcano) compared to flat ground. The black line shows predicted insertion loss from the Yamamoto and Takagi [1992] fit to the Maekawa [1968] empirical chart (Equation 3) while markers show modeled insertion loss (IL_m; Equation 5) from numerical simulations with receivers spaced at 100 m intervals. The grey line shows the predicted insertion loss according to Fehr [1951] (Equation 4). [B] Variation in Fresnel number with range for each lower boundary type. Colors and markers are the same as in Figure 3A.

distance from the center of the source (1396 meters) compared to the 5.5 m distance between the ground and the receiver, making negligible the interference of ground-reflected waves.

Figure 4 compares waveforms and power spectral density (PSD) estimates for the 2D cylindrical and 3D Cartesian simulations described above. In the 2D cylindrical cases the waveforms exhibit bipolar main pulses with relatively symmetrical compression and rarefaction phases (Figure 4A), while the PSDs feature broad peaks around 0.4 Hz (Figures 4C, 4D). Waveforms from 2D cylindrical simulations with topography are generally time-delayed and reduced in amplitude compared to the simulations with flat ground (Figure 4A). In contrast, the 3D Cartesian simulations feature strongly asymmetrical waveforms with single main compression phases (Figure 4B) and nearly white power spectra (Figures 4E, 4F). The differences in waveform shape can be largely attributed to the different source functions. The 2D axisymmetric models uses a spatially distributed Gaussian pulse at time zero, while the 3D model uses a time-varying function (Blackman-Harris window) at a point-source.

Figure 5 compares the distribution of peak amplitudes and modeled attenuation (IL_m ; Equation 5) for the 2D cylindrical and 3D Cartesian synthetic cases described above. In both the 2D and 3D cases, peak am-

plitudes for the flat-ground approximations decay linearly with distance from the source function receiver, as expected for spherical spreading (Figure 5A and 5B, respectively). When topography is included, the amplitudes are reduced at each station in the 2D case (Figure 5A), whereas in the 3D case, the amplitudes are increased at ARI, KOM, and KUR (Figure 5B). These differences are reflected in the IL_m results (Figure 5C), which feature positive attenuation at every station in the 2D cases but only at HAR and SVO in the 3D cases. However, both modeling approaches produce the same relative amplitude and distribution across the network (e.g. ARI is always more attenuated than KUR and less attenuated than HAR). The attenuation predicted by IL_{pY} is clearly greater than the IL_m results from both 2D and 3D modeling.

While both the 2D cylindrical and 3D Cartesian models present validated solutions to the wave equation [de Groot-Hedlin 2017; Fee et al. 2017b; Iezzi et al. 2019b; Kim et al. 2015; Kim and Lees 2014; Maher et al. 2020], several differences between the methods may account for the larger amplitude reductions in the 2D topography simulations compared to the 3D simulations. In the 2D model the source function is a spatially distributed Gaussian pulse, whereas in the 3D case the source is a time-varying Blackman-harris window at a single grid node. These different sources clearly pro-



Figure 4: Synthetic waveforms and spectra for comparable 2D cylindrical and 3D Cartesian modeling methods. [A] Synthetic waveforms from 2D modeling with topography (color) and flat ground (grey scale). Amplitudes are normalized for ease of visualization; the source function (SRC) is normalized by the maximum at SRC while all other waveforms are normalized by the maximum at ARI with flat ground. [B] Synthetic waveforms from 3D modeling with topography [color; Kim and Lees 2014] and with a free field (grey scale). Free field values are multiplied by two to approximate flat ground. Amplitudes are normalized in the same manner as Figure 4A. Multitaper PSD estimates are shown for synthetic waveforms from [C] 2D flat ground simulations; [D] 2D topography simulations; [E] 3D free field simulations; [F] 3D topography simulations. Grey shaded regions indicate frequencies above 1 Hz (maximum source frequency) for which results may be inaccurate.



Figure 5: [A] Peak pressure of 2D synthetic waveforms at Sakurajima station distances for lower boundaries with topography (circles) and flat ground (squares). The orange square represents the approximate source pressure as recorded 10 m outside the boundary of the source region. [B] Peak pressure of 3D synthetic waveforms over Sakurajima topography (circles) [Kim and Lees 2014] and through the free field (FF, squares). Free field values are multiplied by two to approximate flat ground propagation. [C] Modeled insertion loss (IL_m , Equation 5) compared to predicted insertion loss according to Yamamoto and Takagi [1992] (IL_{pY} , Equation 3) and Fehr [1951] (IL_{pF} , Equation 4) for the 2D peak pressures (hexagons) and 3D peak pressures (diamonds) shown in Figure 5A and Figure 5B, respectively.

duce wavefields with different frequency content as exemplified by the 0.4 Hz peak in the 2D flat synthetics (Figure 4C) in comparison to the nearly white spectrum in the 3D free field synthetics (Figure 4E). Since attenuation increases with frequency (see Section 7), we speculate that the less attenuated low-frequency components (<0.5 Hz) in the 3D model may contribute to the heightened amplitudes when compared to the 2D model. Our comparison between 2D and 3D simulations is inherently limited by these different source functions; however, adapting the methods to achieve equivalent source implementations is outside the scope of this study.

Additionally, the 2D model operates in an axisymmetric geometry, assuming azimuthal symmetry in topography around the source-receiver plane, whereas the 3D model uses a Cartesian geometry that allows for more realistic topography. Wavefield interactions with topography outside the 2D azimuthal plane may account for some discrepancies between the methods. For example, the line-of-sight propagation path for KUR follows a drainage channel (Figure 1) that may allow for constructive interference of reflections from the valley walls. This presumably increases recorded amplitudes at KUR compared to both spherical spreading over flat ground and to the 2D model, which does not account for the channel. Ray tracing [e.g. Blom 2020; Blom and Waxler 2012; Fee et al. 2013; Green et al. 2012] could potentially be used to investigate the influence of offpath reflections on the waveform; however, this is outside the scope of this study. Regardless, the preliminary comparison presented here corroborates the main finding of our 2D simulations, which is that attenuation of infrasound at Sakurajima is overpredicted by the screen diffraction approximation.

7 FREQUENCY DEPENDENCE OF ATTENUATION

While Equations 2 and 5 estimate attenuation for time domain amplitudes at fixed wavelengths (e.g. 340 m in Section 5 and 349 m in Section 6) the thin screen approximation also enables analysis of the frequency-dependence of attenuation. We hypothesize that, if the thin screen approximation holds true for volcano infrasound, attenuation of 2D and 3D synthetics for Sakurajima topography will consistently increase with frequency to match the logarithmic relationship proposed by Maekawa [1968]. We investigate this aspect by first defining a new frequency-dependent variable for the Fresnel number (\hat{N}) , where the wavelength in Equation 1 is written as a function of ambient sound speed (c_0) and a frequency vector (f):

$$\hat{N} = \frac{2(R_{\rm t} - R_{\rm d})}{c_0/f}.$$
(6)

This reformulated Fresnel number can be substitued into Equation 3 to express the empirical relation of

ee Presses universitaires de Strasbourg

Maekawa [1968] in the frequency domain (\widehat{IL}_p) :

$$\widehat{IL}_{\rm pY} = \begin{cases} 5+8|\hat{N}|^{0.438} & \text{for } -0.3 < \hat{N} < 1\\ 10\log_{10}\left(\hat{N}\right) + 13 & \text{for } \hat{N} \ge 1. \end{cases}$$
(7)

Finally, the predictions can be compared to estimates from numerical modeling (\widehat{IL}_m) by substituting the waveform amplitudes in Equation 5 for power spectra:

$$\widehat{IL}_m = -10\log\left(\frac{\hat{p}_s}{\hat{p}_r}\right),\tag{8}$$

where \hat{p}_s is a PSD estimate (unit Pa²Hz⁻¹) for a synthetic waveform from a simulation with topography at the lower boundary, and \hat{p}_r is PSD for a synthetic waveform from a simulation with a flat lower boundary.

We examine the frequency dependence of predicted attenuation for five Sakurajima stations and modeled attenuation for the 2D and 3D simulations with Sakurajima topography described in Section 6. Figure 6 shows results for each station. As expected, IL_p increases as a logarithmic function of frequency, with y-intercepts controlled by \tilde{N} . Predicted attenuation increases by approximately 5 dB from 0.1 to 1 Hz, about half an order of magnitude in units of Pa^2Hz^{-1} . Similarly, \widehat{IL}_m for the 2D simulations generally increases with frequency; however, the values are lower than predicted by IL_p , and the increases are smaller (e.g. 2 dB increase from 0.1 to 1 Hz at SVO). Values of IL_m for 3D simulations are also lower than IL_p ; however, different patterns are observed for stations with relatively low \hat{N} (ARI, KUR and KOM) than for stations with higher \hat{N} (HAR and SVO). For example, at KUR, 3D \widehat{IL}_m decreases with frequency and becomes negative above 0.1 Hz (Figure 6C), whereas at SVO the 3D IL_m is relatively constant at approximately 3 dB below 1 Hz (Figure 6E). Artificial increases of high-frequency energy in finite-difference models may be produced by spurious reflections from the corners in the stair-step topography [e.g. de Groot-Hedlin 2004]; however, this effect may be considered negligible at frequencies for which topography is finely discretized. We ran additional 2D and 3D simulations with finer discretization (5 m grid spacing) and observed no changes in the frequency range of interest $(\sim 0.1-1 \text{ Hz})$. We therefore interpret the decreased attenuation with frequency in the 3D model as a result of wavefield interactions with topography outside the source-receiver plane, such that they are not captured in the 2D model. This finding provides further evidence for the influence of full wavefield effects in counterbalancing power losses by diffraction.

The results of this frequency-dependent analysis provide further evidence to suggest that the thin screen approximation overestimates the diffraction losses for volcano infrasound. However, several limitations are inherent to this approach. The synthetic spectra are highly dependent on the source function, which differs



Figure 6: [A]-[E] PSD curves for synthetics from 2D simulations with flat ground (grey) and Sakurajima topography (orange) at the lower boundaries. Each row represents a Sakurajima station as indicated by text at far right (from top to bottom, these are ARI, HAR, KUR, KOM and SVO). [F]-[J] PSD curves for synthetics from 3D simulations representing flat ground (grey lines) and Sakurajima topography (blue lines) at the lower boundaries. In this case the flat ground is approximated by doubling the pressure amplitudes from a free-field simulation. [K]-[O] Frequency dependence of attenuation for Sakurajima stations according to predicted insertion loss (Equation 7; black lines), 2D numerical modeling (Equation 8; orange lines) and 3D numerical modeling (Equation 8; blue lines).

significantly between the 2D and 3D methods (see Figure 4), and may be changed to model different source types [e.g. Iezzi et al. 2019b; Kim et al. 2012]. Furthermore, the approach cannot distinguish between the effect of diffraction and other wavefield-topography interactions such as reflections and scattering, which may also vary with frequency [e.g. Embleton 1996; Pierce 1981] and with topography outside the 2D azimuthal plane. Numerical artifacts may also influence the spectra. For example, spurious reflections may occur from stair-step boundaries in the discretized topography [e.g. de Groot-Hedlin 2004]. However, we consider this effect negligible in our case since we use a relatively fine grid spacing of 30 nodes per wavelength at 1 Hz, in comparison to the minimum of 10 nodes per wavelength [Taflove and Hagness 2005]. Further work to investigate these numerical limitations is outside the scope of this study; however, our results for the cases and frequencies considered suggest that attenuation by diffraction over topography is less than predicted by Maekawa's empirical relationship.

8 Comparison of Synthetic Results to Observed Data

In this study we primarily investigate numerical simulations because they allow direct estimation of wavefield changes for different lower boundary conditions;

Presses universitaires de Strasbourg

however, the ultimate goal of evaluating the thin screen approximation is to rapidly estimate attenuation in observed signals. Here we assess the applicability of our results to 30 eruption events recorded by five stations at Sakurajima during 18-26 July, 2013 [Fee et al. 2014]. These events represent a subset of 74 short-term average/long-term average (STA/LTA) detections made by Matoza et al. [2014]. Events in this subset feature high signal-to-noise ratios (SNR) as evidenced by three or more stations with power spectra above the median of network-averaged noise spectra in the 0.1- to 10-Hz frequency band [e.g. Figure 7B; details in Maher et al. 2020]. The signals vary in waveform character from sustained low-amplitude jet noise to high-amplitude explosions (up to 449 Pa at 2.4 km) with asymmetrical shock-like waveforms [Fee et al. 2014; Matoza et al. 2014]. To accurately reproduce the variety of observed signal types would require multiple modifications to the synthetic source functions, which is outside the scope of this study; however we observe a general agreement between synthetic and observed data as described here.

Figure 7 shows an example of one event out of 30 and features an explosion followed by lower amplitude oscillations. The waveform shapes lie in between the symmetry of the 2D synthetics (Figure 4A) and the strong asymmetry of the 3D synthetics (Figure 4B). The shape of the power spectra agree with synthetics in that a broad peak is present below 1 Hz (Figure 7B). The ob-

served peak pressures (squares) are overpredicted by the 1/r relationship at each station, reflecting anomalously high amplitude at KUR. The amplitude decay for this event is also compared to the 2D and 3D simulation results, where the synthetics are scaled up to match the observation at KUR (circles and diamonds in Figure 7C). The 2D synthetic peak pressures follow 1/rat KOM and SVO while the 3D synthetics are reduced, more closely matching the observations.

While Figure 7 exemplifies one event in the dataset, the amplitudes of all the events are summarized in Figure 8A. The boxplots show the distribution of observed peak amplitudes normalized to the amplitudes at KUR (p_{KUR}), while markers show the corresponding values for 2D synthetics (blue circles), 3D synthetics (green diamonds), thin screen predictions (red squares) and spherical spreading predictions (black line). The broad ranges in observed amplitudes reflect variability in source and propagation conditions across the 30 events recorded over an eight day period. These process potentially include variable source directionality [e.g. Iezzi et al. 2019b; Jolly et al. 2017; Matoza et al. 2013], eruption style [e.g. Fee et al. 2014; Matoza et al. 2014], winds [e.g. de Groot-Hedlin 2017; Kim et al. 2018; Lacanna et al. 2014; Sabatini et al. 2016], temperature inversions [e.g. Fee and Garcés 2007], and changes in vent/crater geometry [e.g. Fee et al. 2017a; Ortiz et al. 2018]. Our synthetics and predictions therefore do not provide a direct representation of each individual event; however, the general agreement in relative amplitude distribution validates our methodologies against the observations.

Figure 8B shows the distributions considering the Fourier amplitude at 0.5 Hz (p_w) rather than peak pressure. Figure 8B may be compared to Figure 3c of Ishii et al. [2020], although the events considered are different, and screen predictions are based on Equations 2 and 5 rather than pressure field calculations. Use of the Fourier amplitude decreases the spread of observed values at each station and more clearly shows the overestimation of ARI, HAR, SVO values by the 1/r prediction.

The values from 3D synthetics fall within the peak pressure range of observations at all stations, while the values from other methods fall within the observation ranges at ARI, KOM, and SVO, but are overpredicted at HAR (largest N). The predicted values according to the screen approximation generally agree with the values obtained by Ishii et al. [2020], although the value at KOM is slightly larger than the 1/r value. This finding shows that our method based on the empirical relationship of Maekawa [1968] gives similar relative amplitudes arrived at by Ishii et al. [2020]; however, we argue that it is misleading to normalize the amplitudes by KUR in this way. By making the values relative to KUR rather than to the source, the normalization obscures the overestimation of attenuation by the screen approximation in the case of volcano topography (Figure 3A). Furthermore, the values obtained according to the screen approximation do not appear to provide an appreciable improvement in matching the observations when compared to 1/r or results from numerical modeling (Figure 8).

9 Discussion

We aim to develop a correction scheme for improved accuracy in volcano-acoustic source parameter estimates by evaluating a rapid method for predicting diffraction losses. A thin screen approximation to diffraction has been proposed in the context of volcano infrasound as an improvement over the 1/r spreading correction alone [Ishii et al. 2020]; however, the predictive ability of the original empirical relationship for diffraction losses [Maekawa 1968] has not been evaluated for volcano infrasound. In this study we use FDTD modeling [de Groot-Hedlin 2017] to show that Maekawa's original empirical relationship (Section 2) successfully predicts power losses due to infrasound diffraction over an idealized kilometer-scale thin screen. This finding validates the scaling of Maekawa's relation to large length scales and infrasonic wavelengths. Furthermore, the empirical relationship yields comparable relative amplitude distributions across the Sakurajima network to the approach taken by Ishii et al. [2020]. However, the absolute amplitudes predicted by the screen approximation overestimate the attenuation in the case of realistic volcano topography. We interpret this finding as a consequence of reflections from topography counteracting losses from diffraction.

9.1 Application of the thin screen approximation to infrasonic wavelengths

The IL_m results from our scaled thin screen simulation (Figure 3A) agree well with the empirical relationship of Maekawa [1968], indicating that Equation 3 is a reasonable approximation for infrasonic wavelengths recorded on the ground in the case of a kilometerscale thin screen. Furthermore, this agreement suggests that the 2D axisymmetric geometry of our numerical model space is adequate to describe losses for a thin screen. Moreover, the insertion loss results for a wedge and wide barrier generally agree with the empirical prediction (<5 dB difference), suggesting that the screen approximation gives a reasonable first-order estimate of infrasound diffraction losses for a variety of simple long barrier types when receivers are on the ground. Equation 3 therefore provides a rapid method to estimate acoustic power losses for infrasound diffraction over simple barriers without resort to time- or computationally-intensive numerical simulations.



9.2 Implications for modeling diffraction of volcano infrasound

The IL_m results from our 2D simulations with Sakurajima Volcano topography are systematically smaller than predicted by IL_{pY} , with differences on order of 10^1 dB (Figure 3). These findings are corroborated by 3D simulations with Sakurajima topography, which feature even smaller $IL_{\rm m}$ than the 2D results (Figure 5C). In the 3D simulations IL_m is negative at stations ARI, KUR, and KOM, indicating gains in amplitudes relative to flat ground propagation (Figure 5C). This suggests that this predictive empirical relationship is not adequate to describe losses to diffraction over complex volcano topography. The losses are better predicted by a different relation proposed by Fehr [1951] and represented by Equation 4, as shown by grey lines in Figures 3A and 5C. This relation is relevant in that it is meant for scenarios with sources and receivers on the ground; however, it is not very useful in that it is given no justification by Fehr [1951], and is only applicable at $N \ge 0.1$, whereas volcano infrasound recording Figure 7: [A] Example of observed waveforms for one event at Sakurajima at 23:06:56.960 on 07/20/2013 (UTC). Amplitudes are normalized by the maximum at ARI (356 Pa) for ease of plot-[B] Multitaper PSD estiting. mates for the waveforms in Figure 7A (original waveform amplitudes). The grey shaded region indicates the noise range across the network as defined by the 5th and 95th percentiles of networkaveraged PSD curves for 50 % overlapping hourly time-windows during the deployment [Maher et al. 2020]. The dashed grey line is the median noise condition (50th percentile). [C] Peak pressure vs distance for the waveforms in Figure 7A (squares). The peak pressures for 2D simulations with topography (circles), 3D simulations with topography (diamonds). The synthetic and predicted amplitudes are scaled to match the observed KUR amplitude. Black line represents expected amplitude decay from spherical spreading based on KUR amplitude. This example is one out of 30 events used to construct the boxplots in Figure 8.

geometries may have lower Fresnel numbers.

Our conclusions contrast with those of Ishii et al. [2020], who suggest that the observed relative amplitude distribution at Sakurajima is better modeled when accounting for the effect of a thin screen than when only geometrical spreading is considered. We found comparable relative amplitude distributions to Ishii et al. [2020] using our method (Figure 8); however we consider it misleading to normalize the screenpredicted amplitudes relative to station KUR in this way. Our synthetics show that attenuation at KUR is nearly zero or even negative (Figure 5C), whereas the screen approximation would predict 8 dB attenuation for the corresponding Fresnel number. Direct use of the Maekawa [1968] empirical relationship results in overprediction of losses to diffraction in the case of volcano infrasound. Consequently we consider the screen approximation inappropriate, even if normalization by KUR causes the relative amplitude distribution to be comparable to observations.

Furthermore, while Ishii et al. [2020] claim that the predictions from the screen approximation offer an improvement over predictions based on 1/r spreading for



Figure 8: [A] Comparison of relative peak pressure values between observations and results from all methods (2D modeling, 3D modeling, screen predictions, and spherical spreading). Values are normalized by peak pressure at KUR to show relative amplitude distributions across the network. Histograms show the distribution of observed values at each station in bins of 0.1 p/p_{KUR} . [B] As for Figure 8A but using the Fourier amplitude at 0.5 Hz (p_w) instead of peak pressure. Note that vertical gray lines indicate zero counts in the histogram bins.

31 events in 2017, we do not see an improvement for 30 events in 2014 (Figure 8). They used the standard deviation distance ($S = |X - \mu|/\sigma$, where X is the predicted amplitude, μ is the mean of the observations and σ is the standard deviation of the observations) to show that S values are generally lower for the screen prediction than 1/r. We calculated S for the observations in Figure 8 and found larger values for the screen method than the 1/r method at stations ARI, KOM, and SVO (Table 1). While this analysis is clearly limited by small sample size (30 events) and the use of only four stations, it illustrates the lack of appreciable improvement of the screen method compared to 1/r.

We conclude that the primary cause of relatively small IL_m values for volcano topography is the focusing of the acoustic wavefield by multiple reflections down concave-upward slopes (e.g. Figure 2). Focusing has been previously postulated at Sakurajima to explain amplitude variations with azimuth [e.g. Kim and Lees 2014; Lacanna et al. 2014]. Ishii et al. [2020] also observed that amplitudes from 31 Vulcanian explosions during August to September 2017 were consis-

Table 1: Standard deviation distances (S) for predictions from the screen approximation and 1/r spreading.

Stations	Screen	1/ <i>r</i>
ARI	0.33	0.11
HAR	2.43	5.10
KOM	0.57	0.01
SVO	1.71	1.19

tently higher than expected from spherical spreading at one station (JIG). Notably, Ishii et al. [2020] used KUR as a reference station to estimate relative amplitude distrubtions based on 1/r, yet the topographic profile for KUR has one of the most concave slopes in the network. Out of the stations considered in this study, KUR also has the lowest IL_m values, indicating the least attenuation losses to diffraction. Acoustic focusing at this station may explain why the KUR-relative amplitudes are larger than observed amplitudes at several stations including ARI, HAR, and SVO. We interpret that the concavity of the slope facilitates constructive interference of reflected waves which counteract amplitude losses from diffraction.

9.3 Limitations of our methodology

Our results suggest that the diffraction of infrasound over volcano topography produces significantly less attenuation than predicted for a thin screen with equivalent Fresnel number; however, several limitations in our methodology must be acknowledged.

Firstly, our numerical simulations are limited in their treatment of atmospheric propagation effects which may increase or decrease power losses depending on the conditions. We use a homogeneous atmosphere to isolate the effect of diffraction and allow direct comparison to the empirical relationship of Equation 3, but in real outdoor propagation the sound speed generally decreases with altitude [Embleton 1996; Pierce 1981]. This vertical gradient enhances upward refraction of the wavefield and leads to reduced acoustic power at the receivers (greater IL_m) compared to the homogeneous atmosphere case. Lacanna et al. [2014] showed that wind and atmopsheric temperature gradients contributed to spatial and temporal deviations from 1/ramplitude decay at Sakurajima over a 21-month period. Additionally, atmospheric variability can change the propagation conditions over short time periods [e.g. Chunchuzov et al. 2011; de Groot-Hedlin 2017; Fee and Garcés 2007; Green et al. 2012; Iezzi et al. 2018; Johnson et al. 2012; Matoza et al. 2009; Ortiz et al. 2018; Sanderson et al. 2020]. For example, changes in atmospheric conditions in the boundary layer (lowermost \sim 3 km of troposphere) caused up to \sim 10 dB pressure differences in recordings of controlled chemical explosions at local distances [<10 km; Kim et al. 2018]. Consequently, our comparison between synthetic and predicted amplitudes to observations (Figure 8) is limited by the unmodeled atmospheric variability.

Secondly, the rigid topography used in our simulations may introduce minor variations in the synthetic waveforms compared to the observations. Real volcano topography can feature ground of finite-impedance (e.g. loose volcanic tephra layers) that may absorb some acoustic energy or convert it to seismo-acoustic coupled waves [e.g. Matoza et al. 2009] and consequently increase IL_m values. A heterogenous distribution of impedance around the volcano could contribute to variations in the relative amplitudes across the network. This effect is commonly assumed negligible for infrasonic wavelengths and <20 km path lengths [Matoza et al. 2019], but warrants further consideration since volcanic landscapes often feature loosely-consolidated near-surface materials (e.g. tephra layers or soil). Additionally, our simulations feature stair-step topography which may lead to spurious reflections and artificial diffraction side-lobes [de Groot-Hedlin 2017]; however, these are much smaller in amplitude than the compressional first arrivals that we use to calculate IL_m . We also ran additional simulations with finer discretization (5 m grid spacing in both the 2D and 3D models) and observed no changes in the frequency range of interest $(\sim 0.1-1 \text{ Hz})$, indicating negligible influence of artifacts arising from topography discretization.

Finally, we acknowledge that the 2013 infrasound network at Sakurajima is not an optimal configuration for thorough diffraction analysis, since there are only five receivers and they are distributed azimuthally rather than radially. We attempted to improve this in our simulations by placing synthetic receivers at 100 m intervals on the topographic profiles (Figure 2) such that we are able to represent a range of Fresnel numbers (Figure 3). However, for comparison to observations we are limited to the five stations and 30 events (Figure 8). Further investigation of the thin screen diffraction approximation should target a volcano with more events and stations, preferably with a radial line configuration, such as at Yasur Volcano [Jolly et al. 2017; Matoza et al. 2017]. Our focus on Sakurajima has allowed di-

BY

rect comparison to the results of Ishii et al. [2020] in addition to previous studies that have addressed topographic effects at this volcano [e.g. Kim and Lees 2014; Lacanna et al. 2014; McKee et al. 2014; Yokoo et al. 2014].

9.4 Directions for future work

Our results suggest that the thin screen diffraction approximation is inadequeate to describe power losses for volcano infrasound, providing a first step towards the development of an analytical method for this purpose. Computationally-intensive numerical simulations remain the most accurate tool for estimating these losses, but future research could work towards a more rapid predictive relationship. For example, we only considered power losses over five topographic profiles at Sakurajima Volcano, but a larger simulation set involving multiple volanoes and azimuths could be used to derive a new semi-empirical relationship between N and *IL*_m. Such an analysis could incorporate regional (15-250 km) to remote (>250 km) recording distances to generalize the relationship for applications beyond local deployments. We note that in addition to N, it may also be useful to consider some measure of topographic roughness [e.g. 2D Fourier power spectral analysis of topography; Perron et al. 2008; Richardson and Karlstrom 2019], since multiple small barriers could reduce recorded acoustic power without increasing the Fresnel number.

Finally, our conclusions draw attention to the significance of acoustic focusing during propagation down concave slopes. This phenomenon has been previously postulated at Sakurajima [e.g. Lacanna et al. 2014] and seems to counteract the losses due to diffraction. The focusing effect could be further investigated with a geometrical acoustics approach such as ray tracing [e.g. Blom and Waxler 2012; Fee et al. 2013; Green et al. 2012]. A comprehensive correction scheme for topographic effects should account for the most significant processes; however, the relative contributions from diffraction, focusing, scattering, and reflections are currently unclear. Further analysis of these effects is required to achieve a rapid correction scheme to account for topography without costly numerical simulations.

10 Conclusions

We used synthetic pressure results from a 2D FDTD method for linear infrasound propagation in an axisymmetric geometry to numerically model the acoustic power losses to diffraction over kilometer-scale barriers including a thin screen, a wedge, a wide barrier, and Sakurajima Volcano topography. We found that the modeled acoustic power losses for the kilometer-scale screen are well-predicted by an empirically-derived relationship [Maekawa 1968] for diffraction of audible sound (20-20,000 Hz) over thin meter-scale screens, thus affirming the applicability of the relationship to infrasonic wavelengths. The power losses are also well predicted to first order for diffraction over the wedge and wide barrier, but the empirical relationship overpredicts losses for Sakurajima topography by $\sim 10^1$ dB. We corroborate these results by comparison to 3D simulations with Sakurajima topography [Kim and Lees 2014], which also feature smaller attenuation values than predicted by Maekawa's empirical relationship. The relative amplitude distributions for 2D synthetics, 3D synthetics, and thin screen predictions generally agree with data from 30 eruption events at Sakurajima, but the screen predictions do not provide an appreciable improvement over the commonly used correction for geometrical spreading.

We conclude that the thin screen approximation proposed by Maekawa [1968] is an inappropriate model for diffraction by volcano topography because attenuation is overestimated in the topography case. This conclusion contrasts with that of Ishii et al. [2020], who conclude that their implementation of a thin screen approximation improves predictions of relative amplitude distributions at Sakurajima compared to geometrical spreading alone. We hypothesize that the true attenuation is smaller than predicted by Maekawa's empirical relationship because constructive interference of reflections down the concave volcano slopes (focusing) causes amplitude increases that counteract losses by diffraction.

Full wavefield numerical simulations such as finitedifference methods remain the most appropriate approach to account for topography [Iezzi et al. 2019a; Kim et al. 2015; Kim and Lees 2014; Lacanna et al. 2014; Lacanna and Ripepe 2013; Maher et al. 2020]. However, an analytical or empirically-derived method that depends only on readily-available topographic information is desirable for rapid interpretations of infrasound data in a rapid deployment or network planning scenario. Future work towards this goal could benefit from the development of a new semi-empirical relationship based on a large suite of numerical simulations with numerous volcanoes and a variety of recording distances.

Acknowledgements

We thank Professor Tracianne Neilsen for helpful discussions in the preparation of this manuscript. The digital elevation models for Sakurajima Volcano were provided by the Japanese Ministry of Land, Infrastructure, Transport and Tourism. This research was supported by National Science Foundation (NSF) grants EAR-1620576 and EAR-1847736.

Author contributions

Maher is a graduate student under the supervision of Matoza. Maher and Matoza formulated the work and experimental design in consultation with de Groot-Hedlin and Gee. Maher ran the 2D simulations and wrote the initial draft of the manuscript. Kim performed the 3D Cartesian propagation simulations. All authors provided input and feedback throughout the research and manuscript preparation.

DATA AVAILABILITY

Infrasound data from the 2013 Sakurajima deployment [network code YO; Fee et al. 2014] are publicly available through the Data Management Center of the Incorporated Research Institutions for Seismology (https: //www.fdsn.org/networks/detail/YO_2013/).

COPYRIGHT NOTICE

© The Author(s) 2021. This article is distributed under the terms of the Creative Commons Attribution 4.0 International License, which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made.

References

- Aizawa, K., C. Cimarelli, M. A. Alatorre-Ibargüengoitia, A. Yokoo, D. B. Dingwell, and M. Iguchi (2016). "Physical properties of volcanic lightning: Constraints from magnetotelluric and video observations at Sakurajima volcano, Japan". *Earth and Planetary Science Letters* 444, pp. 45–55. DOI: 10.1016/j.epsl.2016.03.024.
- Allstadt, K. E., R. S. Matoza, A. B. Lockhart, S. C. Moran, J. Caplan-Auerbach, M. M. Haney, W. A. Thelen, and S. D. Malone (2018). "Seismic and acoustic signatures of surficial mass movements at volcanoes". *Journal of Volcanology and Geothermal Research* 364, pp. 76–106. DOI: 10.1016/j.jvolgeores.2018.09.007.
- Aramaki, S. (1984). "Formation of the Aira caldera, Southern Kyushu, ~22,000 Years Ago". Journal of Geophysical Research 89 (B10), pp. 8485–8501. DOI: 10. 1029/JB089iB10p08485.
- Berenger, J.-P. (1994). "A Perfectly Matched Layer for the Absorption of Electromagnetic Waves". Journal of Computational Physics 200, pp. 185–200. DOI: 10. 1006/jcph.1994.1159.
- Blom, P. (2020). "The influence of irregular terrain on infrasonic propagation in the troposphere". The



Journal of the Acoustical Society of America 148 (4), рр. 1984–1997. дог: 10.1121/10.0002128.

- Blom, P. and R. Waxler (2012). "Impulse propagation in the nocturnal boundary layer: Analysis of the geometric component". The Journal of the Acoustical Society of America 131 (5), pp. 3680-3690. DOI: 10.1121/ 1.3699174.
- Chunchuzov, I. P., S. N. Kulichkov, O. E. Popov, R. Waxler, and J. Assink (2011). "Infrasound scattering from atmospheric anisotropic inhomogeneities". Izvestiya - Atmospheric and Ocean Physics 47 (5), рр. 540-557. дог: 10.1134/S0001433811050045.
- Cimarelli, C., M. A. Alatorre-Ibargüengoitia, K. Aizawa, A. Yokoo, A. Díaz-Marina, M. Iguchi, and D. B. Dingwell (2016). "Multiparametric observation of volcanic lightning: Sakurajima Volcano, Japan". Geophysical Research Letters 43 (9), pp. 4221–4228. DOI: 10.1002/2015GL067445.
- De Angelis, S., A. Diaz-Moreno, and L. Zuccarello (2019). "Recent developments and applications of acoustic infrasound to monitor volcanic emissions". Remote Sensing 11 (11), pp. 1–18. DOI: 10.3390/ rs11111302.
- De Groot-Hedlin, C. D. (2004). "Criteria for discretization of seafloor bathymetry when using a stairstep approximation: Application to computation of Tphase seismograms". The Journal of the Acoustical So*ciety of America* 115 (3), pp. 1103–1113. DOI: 10.1121/ 1.1643361.
- "Infrasound propagation in tropospheric - (2017). ducts and acoustic shadow zones". Journal of the Acoustical Society of America 142 (4), pp. 1816–1827. DOI: 10.1121/1.5005889.
- Ekici, I. and H. Bougdah (2003). "A review of research on environmental noise barriers". Building Acoustics 10 (4), pp. 289-323. DOI: 10.1260 / 135101003772776712.
- Embleton, T. F. W. (1996). "Tutorial on sound propagation outdoors". The Journal of the Acoustical Society of Hohenwarter, D. (1990). "Railway noise propagation America 100 (1), pp. 31-48. DOI: 10.1121/1.415879.
- Ettouney, S. M. and F. R. Fricke (1973). "Courtyard acoustics". Applied Acoustics 6 (2), pp. 119–132. DOI: 10.1016/0003-682X(73)90021-2.
- Fee, D. and M. Garcés (2007). "Infrasonic tremor in the diffraction zone". Geophysical Research Letters 34 (16), pp. 1-5. DOI: 10.1029/2007GL030616.
- Fee, D., M. M. Haney, R. S. Matoza, A. R. Van Eaton, P. Cervelli, D. J. Schneider, and A. M. Iezzi (2017a). "Volcanic tremor and plume height hysteresis from Pavlof Volcano, Alaska". Science 355 (6320), pp. 45-48. DOI: 10.1126/science.aah6108.
- Fee, D., P. Izbekov, K. Kim, A. Yokoo, T. Lopez, F. Prata, R. Kazahaya, H. Nakamichi, and M. Iguchi (2017b). "Eruption mass estimation using infrasound waveform inversion and ash and gas measurements : Evaluation at Sakurajima Volcano , Japan". Earth and Planetary Science Letters 480, pp. 42–52. DOI: 10. 1016/j.epsl.2017.09.043.

- Fee, D. and R. S. Matoza (2013). "An overview of volcano infrasound: From hawaiian to plinian, local to global". Journal of Volcanology and Geothermal Research 249, pp. 123–139. DOI: 10.1016/j.jvolgeores. 2012.09.002.
- Fee, D., R. S. Matoza, K. L. Gee, T. B. Neilsen, and D. E. Ogden (2013). "Infrasonic crackle and supersonic jet noise from the eruption of Nabro". Geophysical Research Letters 40 (May), pp. 4199-4203. DOI: 10.1002/ grl.50827.
- Fee, D., A. Yokoo, and J. B. Johnson (2014). "Introduction to an Open Community Infrasound Dataset from the Actively Erupting Sakurajima Volcano, Japan". Seismological Research Letters 85 (6), pp. 1151–1162. DOI: 10.1785/0220140051.
- Fehr, R. (1951). "The Reduction of Industrial Machine Noise". Proceedings of the National Noise Symposium, USA, October, 1951.
- Fujiwara, K., Y. Ando, and Z. Maekawa (1977). "Noise Control by Barriers - Part 1: Noise Reduction by a Thick Barrier". Applied Acoustics 10 (147-159). DOI: 10.1016/0003-682X(77)90022-6.
- Garcés, M., M. Iguchi, K. Ishihara, M. Morrissey, Y. Sudo, and T. Tsutsui (1999). "Infrasonic precursors to a Vulcanian eruption at Sakurajima Volcano, Japan". Geophysical Research Letters 26 (16), pp. 2537–2540. DOI: 10.1029/1998GL005327.
- Green, D. N., R. S. Matoza, J. Vergoz, and A. Le Pichon (2012). "Infrasonic propagation from the 2010 Eyjafjallajkull eruption: Investigating the influence of stratospheric solar tides". Journal of Geophysical Research Atmospheres 117 (21), pp. 1–19. DOI: 10.1029/ 2012JD017988.
- Hadden, W. J. and A. D. Pierce (1981). "Sound diffraction around screens and wedges for arbitrary point source locations". The Journal of the Acoustical Society of America 69 (5), pp. 1266–1276. DOI: 10.1121/1. 385809.
- models". Journal of Sound and Vibration 141 (1), pp. 17-41. doi: 10.1016/0022-460X(90)90510-7.
- Iezzi, A. M., D. Fee, K. Kim, A. D. Jolly, and R. S. Matoza (2019a). "3-D Acoustic Multipole Waveform Inversion at Yasur Volcano, Vanuatu". Journal of Geophysical Research: Solid Earth, 2018JB017073. DOI: 10. 1029/2018JB017073.
- (2019b). "Three-Dimensional Acoustic Multipole _ Waveform Inversion at Yasur Volcano, Vanuatu". Journal of Geophysical Research: Solid Earth 124 (8), рр. 8679-8703. DOI: 10.1029/2018ЈВ017073.
- Iezzi, A. M., H. F. Schwaiger, D. Fee, and M. M. Haney (2018). "Application of an updated atmospheric model to explore volcano infrasound propagation and detection in Alaska". Journal of Volcanology and Geothermal Research 371, pp. 192–205. DOI: 10.1016/ j.jvolgeores.2018.03.009.
- Iguchi, M., T. Tanegyri, Y. Ohta, S. Ueki, and S. Nakao (2013). "Characteristics of Volcanic Activity at Saku-

rajima Volcano's Showa Crater During the Period 2006 to 2011". Bulletin of the Volcanological Society of Japan 58 (1), pp. 115–135. DOI: 10.18940/kazan. 58.1_115.

- Ishihara, K. (1985). "Dynamical analysis of volcanic explosion". *Journal of Geodynamics* 3 (3-4), pp. 327–349. DOI: 10.1016/0264-3707(85)90041-9.
- Ishii, K., A. Yokoo, M. Iguchi, and E. Fujita (2020). "Utilizing the solution of sound diffraction by a thin screen to evaluate infrasound waves attenuated around volcano topography". *Journal of Volcanology and Geothermal Research* 402, p. 106983. DOI: 10. 1016/j.jvolgeores.2020.106983.
- Johnson, J. B., J. Anderson, O. Marcillo, and S. Arrowsmith (2012). "Probing local wind and temperature structure using infrasound from Volcan Villarrica (Chile)". Journal of Geophysical Research Atmospheres 117 (17), pp. 1–16. DOI: 10.1029/2012JD017694.
- Johnson, J. B. and A. J. C. Miller (2014). "Application of the Monopole Source to Quantify Explosive Flux during Vulcanian Explosions at Sakurajima Volcano (Japan)". Seismological Research Letters 85 (6), pp. 1163–1176. DOI: 10.1785/0220140058.
- Johnson, J. B. and M. Ripepe (2011). "Volcano infrasound: A review". Journal of Volcanology and Geothermal Research 206 (3-4), pp. 61–69. DOI: 10.1016/j. jvolgeores.2011.06.006.
- Johnson, J. B., M. C. Ruiz, H. D. Ortiz, L. M. Watson, G. Viracucha, P. Ramon, and M. Almeida (2018a). "Infrasound Tornillos Produced by Volcán Cotopaxi's Deep Crater". Geophysical Research Letters 45 (11), pp. 5436–5444. DOI: 10.1029/2018GL077766.
- Johnson, J. B., L. M. Watson, J. L. Palma, E. M. Dunham, and J. F. Anderson (2018b). "Forecasting the Eruption of an Open-Vent Volcano Using Resonant Infrasound Tones". *Geophysical Research Letters* 45, pp. 2213–2220. DOI: 10.1002/2017GL076506.
- Jolly, A. D., R. S. Matoza, D. Fee, B. M. Kennedy, A. M. Iezzi, R. H. Fitzgerald, A. C. Austin, and R. Johnson (2017). "Capturing the Acoustic Radiation Pattern of Strombolian Eruptions using Infrasound Sensors Aboard a Tethered Aerostat, Yasur Volcano, Vanuatu". *Geophysical Research Letters* 44, pp. 9672–9680. DOI: 10.1002/2017GL074971.
- Jonasson, H. G. (1972). "Sound reduction by barriers on the ground". Journal of Sound and Vibration 22 (1), pp. 113–126. DOI: 10.1016/0022-460X(72)90849-8.
- Kawakatsu, H., T. Ohminato, H. Ito, and Y. Kuwahara (1992). "Broadband Seismic Observation at the Sakurajima Volcano, Japan". *Geophysical Research Letters* 19 (19), pp. 1959–1962. DOI: 10.1002/ange. 201007361.
- Kim, K., D. Fee, A. Yokoo, and J. M. Lees (2015). "Acoustic source inversion to estimate volume flux from volcanic explosions". *Geophysical Research Letters* 42 (13), pp. 5243–5249. DOI: 10.1002 / 2015GL064466.

- Kim, K. and J. M. Lees (2011). "Finite-difference timedomain modeling of transient infrasonic wavefields excited by volcanic explosions". *Geophysical Research Letters* 38 (6), pp. 2–6. DOI: 10.1029/2010GL046615.
- (2014). "Local Volcano Infrasound and Source Localization Investigated by 3D Simulation". Seismological Research Letters 85 (6), pp. 1177–1186. DOI: 10.1785/0220140029.
- Kim, K., J. M. Lees, and M. Ruiz (2012). "Acoustic multipole source model for volcanic explosions and inversion for source parameters". *Geophysical Journal International* 191 (3), pp. 1192–1204. DOI: 10.1111/ j.1365-246X.2012.05696.x.
- Kim, K., A. Rodgers, and D. Seastrand (2018). "Local Infrasound Variability Related to In Situ Atmospheric Observation". *Geophysical Research Letters* 45 (7), pp. 2954–2962. DOI: 10.1002/2018GL077124.
- Lacanna, G., M. Ichihara, M. Iwakuni, M. Takeo, M. Iguchi, and M. Ripepe (2014). "Influence of atmospheric structure and topography on infrasonic wave propagation". Journal of Geophysical Research: Solid Earth 119 (4), pp. 2988–3005. DOI: 10.1002/ 2013JB010827.
- Lacanna, G. and M. Ripepe (2013). "Influence of nearsource volcano topography on the acoustic wavefield and implication for source modeling". *Journal of Volcanology and Geothermal Research* 250, pp. 9–18. DOI: 10.1016/j.jvolgeores.2012.10.005.
- Le Pichon, A., L. Ceranna, and J. Vergoz (2012). "Incorporating numerical modeling into estimates of the detection capability of the IMS infrasound network". *Journal of Geophysical Research Atmospheres* 117 (5), pp. 1–12. DOI: 10.1029/2011JD016670.
- Li, K. M. and H. Y. Wong (2005). "A review of commonly used analytical and empirical formulae for predicting sound diffracted by a thin screen". Applied Acoustics 66 (1), pp. 45–76. DOI: 10.1016/j. apacoust.2004.06.004.
- Macdonald, H. M. (1915). "A class of diffraction problems". *Proceedings of the London Mathematical Society* 2, pp. 410–427. DOI: 10.1112/plms/s2_14.1.410.
- Maekawa, Z. (1968). "Noise reduction by screens". Applied Acoustics 1 (3), pp. 157–173. doi: 10.1016/0003-682X(68)90020-0.
- Maekawa, Z. and S. Osaki (1985). "A simple chart for the estimation of the attenuation by a wedge diffraction". *Applied Acoustics* 18 (5), pp. 355–368. DOI: 10.1016/0003-682X(85)90054-4.
- Maher, S. P., R. S. Matoza, C. D. de Groot-Hedlin, K. L. Gee, D. Fee, and A. Yokoo (2020). "Investigating Spectral Distortion of Local Volcano Infrasound by Nonlinear Propagation at Sakurajima Volcano, Japan". Journal of Geophysical Research: Solid Earth 125 (3), pp. 1–25. DOI: 10.1029/2019JB018284.
- Maryanto, S., M. Iguchi, and T. Tameguri (2008). "Constraints on the source mechanism of harmonic tremors based on seismological, ground deformation, and visual observations at Sakurajima volcano,

Japan". Journal of Volcanology and Geothermal Research 170 (3-4), pp. 198–217. DOI: 10.1016/j. jvolgeores.2007.10.004.

- Matoza, R. S., D. Fee, D. Green, and P. Mialle (2019). "Volcano Infrasound and the International Monitoring System". *Infrasound Monitoring for Atmospheric Studies*. Ed. by A. Le Pichon. Second. Cham: Springer International Publishing. Chap. 33, pp. 1023–1077. ISBN: 978-3-319-75138-2. DOI: 10.1007/978-3-319-75140-5.
- Matoza, R. S., D. Fee, and T. M. Lopez (2014). "Acoustic Characterization of Explosion Complexity at Sakurajima, Karymsky, and Tungurahua Volcanoes". *Seismological Research Letters* 85 (6), pp. 1187–1199. DOI: 10.1785/0220140110.
- Matoza, R. S., D. Fee, T. B. Neilsen, K. L. Gee, and D. E. Ogden (2013). "Aeroacoustics of volcanic jets : Acoustic power estimation and jet velocity dependence". Journal of Geophysical Research: Solid Earth 118 (April), pp. 6269–6284. DOI: 10.1002 / 2013JB010303.
- Matoza, R. S., M. A. Garcés, B. A. Chouet, L. D'Auria, M. A. Hedlin, C. De Groot-Hedlin, and G. P. Waite (2009). "The source of infrasound associated with long-period events at mount St. Helens". Journal of Geophysical Research: Solid Earth 114 (4), pp. 1–38. DOI: 10.1029/2008JB006128.
- Matoza, R. S., A. D. Jolly, D. Fee, R. Johnson, B. Chouet, P. Dawson, G. Kilgour, B. Christenson, E. Garaebiti, A. M. Iezzi, A. C. Austin, B. M. Kennedy, R. H. Fitzgerald, and N. Key (2017). "Seismo-acoustic wavefield of strombolian explosions at Yasure volcano, Vanuatu using a broadband seismo-acoustic network, infrasound arrays, and infrasonic sensors on tethered balloons". *Abstract PE13B-124*. Portland: IAVCEI.
- McKee, K., D. Fee, C. Rowell, and A. Yokoo (2014). "Network-Based Evaluation of the Infrasonic Source Location at Sakurajima Volcano, Japan". Seismological Research Letters 85 (6), pp. 1200–1211. DOI: 10. 1785/0220140119.
- Miwa, T. and A. Toramaru (2013). "Conduit process in vulcanian eruptions at Sakurajima volcano, Japan: Inference from comparison of volcanic ash with pressure wave and seismic data". Bulletin of Volcanology 75 (685), pp. 1–13. DOI: 10.1007/s00445-012-0685-y.
- Morrissey, M., M. Garcés, K. Ishihara, and M. Iguchi (2008). "Analysis of infrasonic and seismic events related to the 1998 Vulcanian eruption at Sakurajima". *Journal of Volcanology and Geothermal Research* 175 (3), pp. 315–324. DOI: 10.1016/j.jvolgeores.2008. 03.008.
- Morse, P. and K. Ingard (1986). *Theoretical acoustics*. Princeton, N.J.: Princeton University Press.
- Muramatsu, D., K. Aizawa, A. Yokoo, M. Iguchi, and T. Tameguri (2018). "Estimation of Vent Radii From Video Recordings and Infrasound Data Analysis: Implications for Vulcanian Eruptions From Sakurajima

Volcano, Japan". *Geophysical Research Letters* 45, pp. 12829–12836. DOI: 10.1029/2018gl079898.

- Ortiz, H. D., J. B. Johnson, P. G. Ramón, and M. C. Ruiz (2018). "Using infrasound waves to monitor tropospheric weather and crater morphology changes at Volcán Tungurahua, Ecuador". Journal of Volcanology and Geothermal Research 349, pp. 205–216. DOI: 10.1016/j.jvolgeores.2017.11.001.
- Ostashev, V. E., D. K. Wilson, L. Liu, D. F. Aldridge, N. P. Symons, and D. Marlin (2005). "Equations for finitedifference, time-domain simulation of sound propagation in moving inhomogeneous media and numerical implementation". *The Journal of the Acoustical Society of America* 117 (2), pp. 503–517. DOI: 10.1121/ 1.1841531.
- Perron, J. T., J. W. Kirchner, and W. E. Dietrich (2008). "Spectral signatures of characteristic spatial scales and nonfractal structure in landscapes". *Journal of Geophysical Research: Earth Surface* 113 (4), pp. 1–14. DOI: 10.1029/2007JF000866.
- Pierce, A. D. (1974). "Diffraction of sound around corners and over wide barriers". *The Journal of the Acoustical Society of America* 55 (5), p. 941. DOI: 10.1121/1. 1914668.
- (1981). Acoustics: An Introduction to Its Physical Principles and Applications. New York, New York: McGraw-Hill.
- Plotkin, K. J., L. C. Sutherland, and B. T. Vu (2009). "Lift-off acoustics predictions for the Ares I launch pad". 15th AIAA/CEAS Aeroacoustics Conference (30th AIAA Aeroacoustics Conference) (May), pp. 11–13. DOI: 10.2514/6.2009-3163.
- Press, W. H., S. A. Teukolsky, W. Vetterling, and B. P. Flannery (1996). *Numerical Recipes in Fortran77*. 2nd. New York: Cambridge University Press, pp. 716–718.
- Purcell, J. B. C. (1957). "Control of Airborne Sound by Barriers". *Noise Control* 3 (4), pp. 20–58. DOI: 10. 1121/1.2369268.
- Richardson, P. and L. Karlstrom (2019). "The multiscale influence of topography on lava flow morphology". Bulletin of Volcanology 81 (4). DOI: 10.1007/ s00445-019-1278-9.
- Sabatini, R., O. Marsden, C. Bailly, and C. Bogey (2016). "A numerical study of nonlinear infrasound propagation in a windy atmosphere". *Journal of the Acoustical Society of America* 140 (1). DOI: 10.1121/1. 4958998.
- Sanderson, R. W., R. S. Matoza, D. Fee, M. M. Haney, and J. J. Lyons (2020). "Remote detection and location of explosive volcanism in Alaska with the Earth-Scope Transportable Array". *Journal of Geophysical Research*. DOI: 10.1029/2019JB018347.
- Scholes, W. E., A. C. Salvidge, and J. W. Sargent (1971). "Field performance of a noise barrier". *Journal of Sound and Vibration* 16 (4), pp. 627–642. DOI: 10. 1016/0022-460X(71)90668-7.
- Smith, C. M., A. R. Van Eaton, S. Charbonnier, S. R. Mc-Nutt, S. A. Behnke, R. J. Thomas, H. E. Edens, and G.

Thompson (2018). "Correlating the electrification of volcanic plumes with ashfall textures at Sakurajima Volcano, Japan". *Earth and Planetary Science Letters* 492, pp. 47–58. DOI: 10.1016/j.epsl.2018.03.052.

- Taflove, A. and S. C. Hagness (2005). *Computational Electrodynamics: The Finite-Difference Time-Domain Method*. Artech House. ISBN: 9781580538329.
- Tameguri, T., M. Iguchi, and K. Ishihara (2002). "Mechanism of Explosive Eruptions from Moment Tensor Analyses of Explosion Earthquakes at Sakurajima Volcano, Japan". Bulletin of the Volcanological Society of Japan 47 (4), pp. 197–215. DOI: 10.18940/kazan. 47.4_197.
- Tatge, R. B. (1973). "Barrier-wall attenuation with a finite-sized source". *The Journal of the Acoustical Society of America* 53 (5), pp. 1317–1319. DOI: 10.1121/1.1913472.
- Uhira, K. and M. Takeo (1994). "The source of explosive eruptions of Sakurajima volcano, Japan". Journal of Geophysical Research 99 (94), pp. 17775–17789. DOI: 10.1029/94JB00990.
- Vu, B. (2007). Predicting the Acoustic Environment Induced by the Launch of the Ares I Vehicle. Tech. rep. KSC Technology Development and Application 2006-2007: National Aeronautics and Space Administration, pp. 36–47.

- Yamamoto, K. and K. Takagi (1992). "Expressions of Maekawa's chart for computation". Applied Acoustics 37 (1), pp. 75–82. DOI: 10.1016/0003-682X(92)90012-H.
- Yokoo, A. and K. Ishihara (2007). "Analysis of pressure waves observed in Sakurajima eruption movies". *Earth, Planets and Space* 59 (3), pp. 177–181. DOI: 10. 1186/BF03352691.
- Yokoo, A., Y. J. Suzuki, and M. Iguchi (2014). "Dual Infrasound Sources from a Vulcanian Eruption of Sakurajima Volcano Inferred from Cross-Array Observation". Seismological Research Letters 85 (6), pp. 1212– 1222. DOI: 10.1785/0220140047.
- Yokoo, A., T. Tameguri, and M. Iguchi (2009). "Swelling of a lava plug associated with a Vulcanian eruption at Sakurajima Volcano, Japan, as revealed by infrasound record: Case study of the eruption on January 2, 2007". Bulletin of Volcanology 71 (6), pp. 619–630. DOI: 10.1007/s00445-008-0247-5.
- Young, S. M., K. L. Gee, T. B. Neilsen, and K. M. Leete (2015). "Outdoor measurements of spherical acoustic shock decay." *The Journal of the Acoustical Society of America* 138 (3), EL305. DOI: 10.1121/1.4929928.